Author's response to reviews

Title: Ultrasonographic detection and assessment of the severity of Crohn's disease recurrence after ileal resection

Authors:

Nadia Pallotta (nadia.pallotta@uniroma1.it)
Maurizio Giovannone (mau.giov@tiscali.it)
Patrizio Pezzotti (pezzotti@asplazio.it)
Alessandro Gigliozzi (a.gigliozzi@asl.rieti.it)
Fausto Barberani (fausto949@gmail.com)
Daria Piacentino (daria.piacentino@gmail.com)
Naima Abdulkadir Hassan (naima_mafuto@hotmail.it)
Giuseppina Vincoli (vincoligiousy@libero.it)
Mauro Tosoni (mauro.tosoni@libero.it)
Alfredo Covotta (alfredo.covotta@uniroma1.it)
Adriana Marcheggiano (loly.clarkemarcheggiano@uniroma1.it)
Mauro Di Camillo (msdicamillo@hotmail.com)
Enrico Corazziari (enrico.corazziari@uniroma1.it)

Version: 4 Date: 25 June 2010

Author's response to reviews:

Rome 25 June, 2010

RE: MS 1903589547354754

Dear Doctor Rachel Neilan,

We submit for publication the revised manuscript conformed to the journal style, that takes into consideration the comments of the Editor.

Major revision: From the tables bullet points have been removed

Minor revision: Title page: running head has been removed; each affiliation has been written in full; Header has been removed; The visible vertical lines have been removed

Typographical errors have been checked and corrected.

On behalf of all contributing Authors,

With best regards

Prof Enrico Corazziari